Author's response to reviews

Title: The facilitating factors and barriers encountered in the adoption of a humanized birth care approach in a highly specialized university affiliated hospital

Authors:

Roxana Behruzi (roxana.behruzi@umontreal.ca)
Marie Hatem (marie.hatem@umontreal.ca)
Lise Goulet (lise.goulet@umontreal.ca)
William Fraser (william.fraser@umontreal.ca)

Version: 6 Date: 15 November 2011

Author's response to reviews: see over
November 14, 2011

Dear Editor

Please find enclosed our final corrections regarding to your last comments. We hope this is satisfactory, and please don’t hesitate to contact me if there is anything else we can help with the paper.

Sincerely
Roxana Behruzi, MSc. PhD
Department of Family Medicine
McGill University
515 Pin Avenue West
Montreal, Qc, H2W 1S4, Canada
Roksana.behruzi@mail.mcgill.ca

Major revisions

Tables: table 1 has been submitted in one part.

Minor revisions

Titles: "Method" has changed to "Methods"

Figure legends: we placed figure, on a portrait page layout instead of landscape.

Typography: We checked the manuscript for any typographical errors.